Author's response to reviews

Title: The Frequency Of Microscopic And Focal Active Colitis In Patients With Irritable Bowel Syndrome

Authors:

Kamil Ozdil (kamilozdil@gmail.com)
Abdurrahman Sahin (arahmansmd@hotmail.com)
Turan Calhan (trncalhan@hotmail.com)
Resul Kahraman (drkahraman@hotmail.com)
Adil Nigdelioğlu (nigdelioglu@yahoo.com)
Umit Akyuz (akyuzfu@hotmail.com)
Hacı M Sokmen (hmsokmen@hotmail.com)

Version: 8 Date: 26 August 2011

Author's response to reviews: see over
We would like to thank the editors comments to improve the quality of the manuscript. We have revised the manuscript in accordance with the editors’ suggestions.

**Comment 1:** Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution. The affiliations 1 and 2 need to be written in full.

**Response 1:** The affiliations were corrected accordance with editors’ suggestions in title page.

**Comment 2:** References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

**Response 2:** The authors were added to the references according to the suggestions.